\documentclass[aspectratio=169]{beamer}
\title{Presentation 3}
\subtitle{Mid-Project Review}
\author{CS460/660 - Machine Learning}
\date{26--27 Oct 2026}
\begin{document}
\begin{frame}{Base paper and problem setup}
\begin{itemize}
  \item Base paper and motivation
  \item What is already solved?
  \item Input and output
  \item Dataset and preprocessing
\end{itemize}
\end{frame}
\begin{frame}{Research question and tentative timeline}
\begin{itemize}
  \item What problem are you asking?
  \item Scope, assumptions, and expected contribution
  \item Tentative deliverables and dates
\end{itemize}
\end{frame}
\begin{frame}{Presentation 2 summary}
\begin{itemize}
  \item Attempts made and reasons for them
  \item Results, failures, and lessons learned
  \item Updated timeline
\end{itemize}
\end{frame}
\begin{frame}{Results}
\begin{itemize}
  \item Current results and comparisons
  \item Evidence: figures, tables, or examples
\end{itemize}
\end{frame}
\begin{frame}{Update and next steps}
\begin{itemize}
  \item What changed since Presentation 2?
  \item Remaining work before the final presentation
\end{itemize}
\end{frame}
\begin{frame}{Limitations and questions}
\begin{itemize}
  \item Current limitations and open questions
  \item Feedback needed for the final phase
\end{itemize}
\end{frame}
\end{document}
